

GOPEN ACCESS

Citation: Falay D, Hardy L, Tanzito J, Lunguya O, Bonebe E, Peeters M, et al. (2022) Urban rats as carriers of invasive *Salmonella* Typhimurium sequence type 313, Kisangani, Democratic Republic of Congo. PLoS Negl Trop Dis 16(9): e0010740. https://doi.org/10.1371/journal. pntd.0010740

Editor: Travis J. Bourret, Creighton University, UNITED STATES

Received: May 25, 2022

Accepted: August 13, 2022

Published: September 6, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010740

Copyright: © 2022 Falay et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files. All genomes obtained by WGS **RESEARCH ARTICLE**

Urban rats as carriers of invasive *Salmonella* Typhimurium sequence type 313, Kisangani, Democratic Republic of Congo

Dadi Falay^{1,2,3}, Liselotte Hardy^{5,6}, Jacques Tanzito⁴, Octavie Lunguya^{5,6}, Edmonde Bonebe⁵, Marjan Peeters³, Wesley Mattheus⁷, Chris Van Geet⁸, Erik Verheyen^{9,10}, Dudu Akaibe⁴, Pionus Katuala⁴, Dauly Ngbonda¹, François-Xavier Weill¹¹, Maria Pardos de la Gandara¹¹, Jan Jacobs^{2,3}

 Department of Pediatrics, University Hospital of Kisangani, Kisangani, the Democratic Republic of the Congo, 2 Department of Microbiology, Immunology and Transplantation, KU Leuven, Leuven, Belgium,
Department of Clinical Sciences, Institute of Tropical Medicine, Antwerp, Belgium, 4 Biodiversity
Monitoring Center (Centre de Surveillance de la Biodiversité, CSB), Faculty of Science, University of Kisangani, Kisangani, the Democratic Republic of the Congo, 5 Department of Medical Biology, National Institute for Biomedical Research, Kinshasa, the Democratic Republic of the Congo, 6 Department of Microbiology, University Teaching Hospital of Kinshasa, Kinshasa, Democratic Republic of the Congo,
Sciensano, Infectious Diseases in Humans, Bacterial Diseases, Brussels, Belgium, 8 Department of Cardiovascular Sciences and Pediatrics, KU Leuven and University Hospital Leuven, Leuven, Belgium, 9 OD Taxonomy & Phylogeny, Royal Belgian Institute of Natural Sciences, Brussels, Belgium, 10 Evolutionary Ecology, University of Antwerp, Antwerp, Belgium, 11 Institut Pasteur, Université Paris Cité, Unité des bactéries pathogènes entériques, Paris, France

* lhardy@itg.be

Abstract

Background

Invasive non-typhoidal *Salmonella* (iNTS–mainly serotypes Enteritidis and Typhimurium) are major causes of bloodstream infections in children in sub-Saharan Africa, but their reservoir remains unknown. We assessed iNTS carriage in rats in an urban setting endemic for iNTS carriage and compared genetic profiles of iNTS from rats with those isolated from humans.

Methodology/Principal findings

From April 2016 to December 2018, rats were trapped in five marketplaces and a slaughterhouse in Kisangani, Democratic Republic of the Congo. After euthanasia, blood, liver, spleen, and rectal content were cultured for *Salmonella*. Genetic relatedness between iNTS from rats and humans—obtained from blood cultures at Kisangani University Hospital—was assessed with multilocus variable-number tandem repeat (VNTR) analysis (MLVA), multilocus sequence typing (MLST) and core-genome MLST (cgMLST). 1650 live-capture traps yielded 566 (34.3%) rats (95.6% *Rattus norvegicus*, 4.4% *Rattus rattus*); 46 (8.1%) of them carried *Salmonella*, of which 13 had more than one serotype. The most common serotypes were II.42:r:- (n = 18 rats), Kapemba (n = 12), Weltevreden and Typhimurium (n = 10, each), and Dublin (n = 8). *Salmonella* Typhimurium belonged to MLST ST19 (n = 7 rats) and the invasive ST313 (n = 3, isolated from deep organs but not from rectal content). Sixteen and analyzed on EnteroBase are available at the ENA depository under project number PRJEB54047.

Funding: JJ received financial support for the study from the Flemish Interuniversity Council (VLIR-UOS, rat sampling) and the Belgian Directorate General of Development Cooperation (blood culture surveillance). XFW received financial support for the microbial genomics analyses from the Institut Pasteur and the French Government's Investissement d'Avenir program, Laboratoire d'Excellence "Integrative Biology of Emerging Infectious Diseases" (grant no. ANR-10-LABX-62-IBEID. D.F. has a PhD scholarship from KU Leuven through the Marc Vervenne fund. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

human *S*. Typhimurium isolates (all ST313) were available for comparison: MLVA and cgMLST revealed two distinct rat-human clusters involving both six human isolates, respectively, *i.e.* in total 12/16 human ST313 isolates. All ST313 Typhimurium isolates from rats and humans clustered with the ST313 Lineage 2 isolates and most were multidrug resistant; the remaining isolates from rats including *S*. Typhimurium ST19 were pan-susceptible.

Conclusion

The present study provides evidence of urban rats as potential reservoirs of *S*. Typhimurium ST313 in an iNTS endemic area in sub-Saharan Africa.

Author summary

Dadi (°1974, DR Congo) is a Medical Doctor (Kisangani University 2005) with a master in pediatrics (Kisangani University 2015) with special interest in infectious diseases and tropical medicine. He has 11 years of field research experience. He was team member of scientific expedition "*Boyekoli Ebale Congo*" in 2010 as medical support for the researchers. He conducted work field in a multidisciplinary framework with biologists from faculty of sciences (university of Kisangani) exploring zoonotic diseases in several places in the Congo. Currently, he is doing his PhD research at KU Leuven and Institute of Tropical Medicine Antwerp (ITM), Belgium. Passionate about transmissible diseases, Dadi is exploring the potential reservoirs of non-typhoidal *Salmonella* in Kisangani, DR Congo.

1 Introduction

Worldwide, invasive non-typhoidal *Salmonella* (iNTS) infections account for 535,000 (95% uncertainty interval 409 000–705 000) invasive disease cases. Sub-Saharan Africa accounts for the largest proportion (79.9%), the highest incidence (34.5% (26.6–45.0)) and a fatality ratio of 15.8% (10.0–22.9%) [1]. Risk factors for iNTS infection in sub-Saharan Africa are age < 5 years old, *Plasmodium falciparum* malaria and HIV-infection [2,3]. Main invasive NTS sero-types are *Salmonella enterica* subspecies *enterica* Typhimurium and Enteritidis (hereafter named *S*. Typhimurium and *S*. Enteritidis), in particular *S*. Typhimurium ST313 [4–7]. This contrasts with high-income countries, where most frequent *S*. Typhimurium sequence type is ST19 [8,9], that causes self-limiting diarrhea with only occasional progression to bloodstream infection [10]. In addition to their virulence, iNTS display high prevalence of antimicrobial resistance [2,11,12].

In high-income countries, the reservoir of non-typhoidal *Salmonella* is zoonotic and transmission is mainly foodborne [10,13,14]. In contrast, iNTS infections in sub-Saharan Africa are believed to be anthroponotic [7,15,16]. In line with *Salmonella* Typhi and Paratyphi A causing enteric fever in humans [17], invasive S. Typhimurium and Enteritidis have genetically adapted to humans [2,11]. These invasive NTS are clinically specific: they are mainly causing bloodstream infections (and not diarrhea) and are associated with high mortality [18,19]. Further, household studies of index patients with NTS bloodstream infections in Kenya and Burkina-Faso have demonstrated NTS stool excretion in household members but not in household livestock [15,20]. However, reviewing the evidence of source attribution, several authors stated that an animal reservoir for NTS in sub-Saharan Africa is not excluded and needs further study [21–23].

The Democratic Republic of Congo (DR Congo) is highly endemic for iNTS disease. The National Institute of Biomedical Research (INRB) in Kinshasa coordinates, in collaboration with the Institute of Tropical Medicine (ITM) in Antwerp, Belgium, a nationwide microbiological surveillance network processing blood cultures in patients suspected of bloodstream infections. Kisangani University Hospital participates—through hospitals and health centers within and around Kisangani—in this network [24]. Over the past decade, NTS consistently represented the most frequent blood culture isolates in children attending the hospitals and health centers in this network, with *S*. Typhimurium and Enteritidis isolates representing respectively 63.1% and 36.4% of iNTS isolates of which 87.4% displaying MDR [19].

As a fast-growing densely populated metropolis, Kisangani provides a wide array of ecological niches attractive for rodents. Previously, Kisangani University had assessed rats trapped in marketplaces in Kisangani as reservoirs of *Rickettsia typhi* and *Bartonella* spp. [25]. Rodents have been described as a reservoir and vector of many pathogens including *Salmonella* [26] but to our knowledge, rats have not yet been assessed as a potential reservoir of NTS. Therefore, we aimed to assess rats present in marketplaces in Kisangani for carriage of NTS. Our objectives were to estimate the proportion of *S*. Typhimurium and Enteritidis carriers (rats with *S*. Typhimurium or Enteritidis isolates from liver, spleen, blood, and/or intestinal content) among rats, and to assess the antimicrobial resistance profile of the *Salmonella* isolates.

2 Materials and methods

2.1 Ethics statement

Ethical approval for the blood culture surveillance study was granted by the Institute of Tropical Medicine Antwerp Institutional Review Board (approval number: 613/08), the Ethics committee of Antwerp University (approval number: 8/20/96) and the Ministry of Health of the Democratic Republic of the Congo (approval number: 8/20/96). For the rat carrier study, authorization by the health authorities of Tshopo Province (Division Provincial de la Santé) was obtained on 12 May 2016 and procedures were carried out according to the 2016 Guidelines of the American Society of Mammologists for the use of wild mammals in research and education [27]. Before starting rat trapping, the responsible in charge of security at the trapping site was contacted for permission and surveillance of the traps.

2.2 Study design

In an exploratory study, rats trapped alive in marketplaces in the Kisangani area were sampled for *Salmonella* in the microbiological laboratory at the University Hospital of Kisangani. *Salmonella* isolates were tested for serotype and antimicrobial susceptibility. S. Typhimurium and Enteritidis isolates were compared by molecular typing for genetic relatedness with clinical isolates obtained from blood cultures.

2.3 Rat sampling

2.3.1 Sample size calculation. Based on CLSI guidelines, a minimum sample size of 30 *Salmonella* spp. isolates was set for reliable data about serotype distribution and antimicrobial susceptibility profiles [28]. At an assumed NTS carrier ratio of 5-10% [29–32] we targeted to capture a total of 600 rats to have sample size of 30 rats carrying *Salmonella*.



Fig 1. Overview of sampling sites in Kisangani city. Kisangani map was made using OpenStreetMap data, Stamen open source maps (<u>https://stamen.com/open-source/</u>) and the R-package ggmap [34].

2.3.2 Rat collection settings. The rat sampling was conducted between April 2016 and December 2018, with interruptions due to administrative and/or logistical issues from January to February 2017 and from April 2017 to April 2018 (S1 Fig). Kisangani counts approximately one million inhabitants; 70 to 80% are living below the poverty line [33]. Sampling sites were selected and described previously [25] (Fig 1). The Central Market is the largest market in Kisangani city, located in the city center, as is the case for Makolo-Ngulu market, Tshopo market and Christ-Roi market. Yaté Market is in the secondary port of Kisangani, at the Congo River. Merchants from the surrounding villages temporarily reside at the market. The Mangobo slaughterhouse, at the bank of the Tshopo river, is situated in a savannah habitat, close to a small port trading cattle imported from the eastern provinces (Ituri and Nord-Kivu). In all these sites, the goods are displayed on tables made of unpolished wood allowing easy access by rats. In some places, the products for sale are placed on the ground (*e.g.* rice or corn put to dry). There is no rodent control policy in the city of Kisangani.

2.3.2 Rat sampling procedure. Tomahawk live-capture traps (Tomahawk Trap Co., Tomahawk, WI, USA) (dimensions 490/178/173 mm) were used with smoked fish or meat as bait. The traps were installed in warehouses, rice mills, or sheltered places in the sampling sites, to prevent them from being stolen or misused. Traps were installed between 5–7 p.m. and collected the next day between 6–8 a.m. Using a global positioning system (GPS) (Garmin 60 Cx Southampton, Hampshire, UK), the location of each positive trap was recorded.

Trapping success was defined as the proportion of rats caught per number of traps installed in the trapping site. Trapped rats were transported to the Faculty of Sciences in Kisangani. At reception, rats were taken out of the trap using a single-use cloth bag for each animal. Still in the bag (to prevent scratches or bites), they were euthanized with an intraperitoneal injection of a lethal dose of ketamine (Ketalin, Shalina laboratories PVT LTD, Mumbai India, 0.5-1ml of 250mg/5ml solution). The delay between collection from the trapping site and euthanasia was less than 3 hours and the animal was dissected immediately after the euthanasia (less than 10 minutes). Rat species (*Rattus norvegicus* versus *Rattus rattus*) was determined on morphological characteristics. Sexual status, length and mass were recorded.

After euthanasia, rats were placed in a dissection tray and the abdomen was sprayed with 3-5ml of 80% ethanol to disinfect the skin. Using a sterile scalpel blade, an incision was made throughout the length of the abdomen. Different samples (2–3 g, each sampled with separate sterile instruments) were collected from the spleen and subsequently from the liver. From May 2018 onwards, additional blood (0.5-1ml) was sampled by intracardiac puncture, before sampling of spleen and liver. At the end of the procedure, the rectal content was collected. Internal organs collected were transported to the lab for preliminary analysis within 4 hours.

The samples of spleen, liver and rectal content were placed in a sterile Petri dish, sliced, and transferred into 10ml Selenite broth (Becton Dickinson and Company, Franklin Lakes, NJ, US) followed by vortex mixing. Blood was inoculated directly into the Selenite broth tube. Selenite broth tubes were transported to the microbiology laboratory within 4 hours after inoculation.

2.4 Salmonella identification, serotyping, and antibiotic susceptibility testing in rats

Laboratory processing was done as described previously for stool samples [35]. Selenite broth tubes were incubated at 35°C for 12–18 hours, whereafter 10 µl broth was inoculated on two *Salmonella-Shigella* (SS) agar plates (Lab M Limited, Lancashire, UK). These were incubated at 35°C and read after 18-24h. In case of no growth, the plates were evaluated after another 24 hours of incubation. In case of growth, up to five (to detect the presence of multiple serop-types) colonies suspected of *Salmonella* were transferred to Kligler Iron Agar (KIA) tubes (Lab M Limited) and incubated for 18–24 hours at 35°C. Bacterial growth in the KIA tubes suggestive of *Salmonella* was biochemically confirmed by a panel of disk-based biochemical tests (DiaTabs, Rosco, Taastrup, Denmark). Isolates with a reaction pattern compatible with *Salmonella* were stored in 2 ml tubes of Trypticase Soy Agar (Oxoid, Basingstoke, UK) and shipped to the Institute of Tropical Medicine (ITM, Antwerp, Belgium) for serotyping and antibiotic susceptibility testing (AST).

Serotyping was done by commercial antisera (Vison, Pro-lab Diagnostics Inc., Richmond Hill, Ontario, Canada). AST was done by disk diffusion (Neo-Sensitabs, Rosco, Taastrup, Denmark) and, in the case of azithromycin and ciprofloxacin, by E-test macromethod (bioMérieux, Marcy L'Etoile, France) for assessment of minimal inhibitory concentration values (MIC-values) as previously described [19] (S3 Table). Results were interpreted according to the Clinical and Laboratory Standards Institute (CLSI) M100-S31 criteria [36]. Multidrug resistance (MDR) was defined as combined resistance to amoxicillin, trimethoprim-sulfamethoxazole, and chloramphenicol [12].

2.5 Human sampling

Salmonella isolates were obtained and stored as part of this INRB-ITM microbiological blood culture surveillance network at Kisangani University Hospital as previously described [24]. Basic clinical and demographic data were recorded. To assess the genetic relatedness between *Salmonella* isolates from rats and humans, blood culture isolates obtained during the rat trapping period and six months before and 6 months after this period (01 October 2015–30 June 2019) were selected. Serotyping and AST of blood culture *Salmonella* isolates were done as described previously [19].

2.6 Whole genome sequencing

For all but one *Salmonella* spp. isolate, whole genome sequencing (WGS) and analysis was performed as previously described [37]. Total DNA was extracted from overnight cultures at 37°C in tryptic soy broth (TSB) (Oxoid) using the MagNAPure 96 system (Roche Diagnostics, Meylan, France). WGS was performed by the genomic platform of the Institut Pasteur, in Paris, France ("Plateforme de microbiologie mutualisée", P2M). The libraries were prepared with the Nextera XT kit (Illumina) and sequencing was performed with the NextSeq 500 system (Illumina, San Diego, CA, USA), generating 150 bp paired-end reads. The short reads were assembled de novo with SPAdes version 3.6.0.23. Serotype prediction, multilocus sequence typing (MLST), and core genome MLST (cgMLST) were performed with various tools integrated into EnteroBase (https://enterobase.warwick.ac.uk/). The presence and type of antimicrobial resistance genes (ARGs) or ARG-containing structures were determined with ResFinder version 4.1 (https://cge.cbs.dtu.dk/services/ResFinder/) and PlasmidFinder version 1.3 (https://cge.cbs.dtu.dk/services/PlasmidFinder/) on SPAdes assemblies.

2.6.1 Genetic relatedness between Salmonella isolates from rats and humans. Genetic relatedness between the iNTS isolates (in case those belonging to *S*. Typhimurium serotype, see <u>Results</u>) was assessed by three different approaches: multilocus variable-number tandem repeat (VNTR) analysis (MLVA), MLST, and cgMLST.

MLVA was performed at Sciensano (Brussels, Belgium) as described previously [35]. MLVA profiles were established based on five loci ordered STTR9-, STTR5-, STTR6-, STTR10-, and STTR3-. Identical *S*. Typhimurium MLVA types (clusters) were defined as isolates with variation in none or one of the rapidly changing loci (STTR5, STTR6, and STTR10) but no variation in the stable loci (STTR3 and STTR9) [38].

For WGS, the 7-genes MLST scheme was applied, based on the sequences of the seven housekeeping genes: *aroC*, *dnaN*, *hemD*, *hisD*, *purE*, *sucA*, *thrA* [39,40]. Hierarchical clustering on cgMLST (HierCC) and minimum spanning trees (MS-trees) was performed on EnteroBase based on the sequences of 3002 loci for *Salmonella* [41]. On the HierCC scheme, HC2850 values allow to deduct the *Salmonella* subspecies (*i.e.* difference of \leq 150 or less alleles from the 3002 loci in the cgMLST scheme); HC200 values provide a similar discrimination as the 7-genes MLST typing (\leq 200 alleles difference between genomes); and HC5 values can be considered a strong indicator of epidemiological relatedness (i.e. if no more than five alleles are different) [41,42].

2.6.2 Genetic relatedness between isolates from this study and the literature. We compared some of the isolates of *S*. Typhimurium and *S*. Enteritidis identified in this study with previous studies on iNTS in Africa [7,43] using EnteroBase. This database recovers regularly *Salmonella* genomes from the GenBank Sequence Read Archive (SRA). The "Custom View" utility allows to add metadata for genomes already present on the site. EnteroBase has recovered 564 genomes from the study on *S*. Typhimurium ST313 by Pulford *et al.* [7] and 503 genomes from the study on *S*. Enteritidis ST11 by Feasey *et al.* [43]. A "Custom View" was created to add the information on lineages and plasmid repertoire for the *S*. Typhimurium ST313 from the Pulford *et al.* study [7], that was next compared to the HierCC clustering on cgMLST by EnteroBase [41]. A "Custom View" previously created, including the hierBAPS clade/cluster data described by Feasey *et al.* [37] was used to compare the hierBAPS clustering with the EnteroBase HierCC clustering. A minimum spanning (MS) tree (MStree V2 or GrapeTree) based on the EnteroBase "cgMLST V2 + HierCC V1" scheme was produced for each serotype to estimate the allelic distances between isolates from this study and the genomes from Pulford *et al.* [7] and Feasey *et al.* [43].

For the *S*. Typhimurium ST313 in this study, four plasmids, previously associated with the main lineages circulating in Africa: pSLT-BT (accession: FN432031.1), pBT1 (accession: LS997975.1), pBT2 (accession: LS997976.1), and pBT3 (accession: LS997977.1) were searched, and this information was included in the *S*. Typhimurium ST313 "Custom View".

For the *S*. Enteritidis ST11 in this study, plasmid pSEN-BT (accession: LN879484.1) described by Feasey *et al.* [43] as characteristic of the hierBAPS clade 9, also denominated Central/Eastern African clade was also searched and the information was included in the *S*. Enteritidis ST11 "Custom View".

	Central Market	Yaté Market	Slaughter house Mangobo	Christ-Roi Market	Makolo-Ngulu Market	Tshopo Market	Total
Outreach visits	14	26	5	2	3	5	55
Traps installed	420	780	150	60	90	150	1650
Rats caught (% of traps)	128 (30.4%)	380 (48.7%)	12 (8.0%)	4 (6.7%)	16 (17.8%)	26 (17.3%)	566 (34.3%)
Salmonella carriers (% of rats)	12 (9.4%)	30 (7.9%)	2 (16.7%)	1 (25.0%)	0 (0.0%)	1 (3.8%)	46 (8.1%)

Table 1. Numbers of outreach	visits, rats caught and Salmon	ella carriers over the six sa	mpling sites in	Kisangani 2016-2018

2.7 Data management and analysis, definitions

Geographic coordinates, morphometric measurements of rats and microbiological data were entered in an Excel database (Microsoft, Redmond, WA, USA). Rats with at least one isolate confirmed as *Salmonella* were categorized as *Salmonella* carriers. Several rats carried more than one *Salmonella* serotype. Only the first isolate per serotype per rat was used for the analysis of the distribution of serotypes and AST. For the display of MLVA typing results, all *Salmonella* Typhimurium isolates were assessed but only the first isolate per MLVA type per rat was presented. All genomes obtained by WGS and analyzed on EnteroBase are available at the ENA depository under project number PRJEB54047 (S1 Table).

3 Results

3.1 Rats sampling

During 55 outreach visits, 566 (34.3%) of 1,650 traps yielded 566 rats including 541 (95.6%) *Rattus norvegicus* (Norwegian or brown rat) and 25 (4.4%) *Rattus rattus* (roof rat or black rat). The highest trapping success was seen in Central and Yaté marketplaces (30.4% and 48.7% respectively). There were no incidents (*e.g.* tripping) with the traps reported and apart from one cat, no other non-rat species was captured. A total of 46 (8.1%) rats carried *Salmonella* spp. (Table 1). All captured rats were alert and active, and none showed signs of illness and weakness. Most rats (89.0%, 504/566) were sexually mature, with a male-to-female ratio of 0.91; median (Interquartile Range, IQR) length (without tail) and mass were 22.1 cm (8.3–28.4 cm) and 280 g (30–515 g) respectively; showing no difference between *Salmonella* carriers and non-carriers (S2 Table).

3.2 Distribution of serotypes and specimens in rats

The 46 *Salmonella* rat carriers yielded a total of 254 *Salmonella* isolates, of which 253 were available for further analysis, one isolate was not viable after transport. Table 2 lists the *Salmonella* serotypes isolated per body specimen and rat. Thirteen rats yielded multiple serotypes and a total of 73 non-duplicate serotypes isolates were recovered. Most common serotypes were serotype II.42:r:- (found in 18 rats), Kapemba (12 rats), Weltevreden and Typhimurium (10 rats each), and Dublin (found in 8 rats). All *Salmonella* Dublin non-duplicate serotype isolates (n = 11) and all but one *S*. Typhimurium isolates (n = 12 in 10 rats) were obtained exclusively from deep organ samples (spleen or liver). *S*. Entertitidis was not isolated from rats.

3.3 Salmonella serotypes obtained from human blood cultures

A total of 40 non-duplicate *Salmonella* isolates were recovered from a total of 2742 blood cultures performed. Two isolates were not viable upon retrieval from shipment and not available for serotyping. Main serotypes comprised Typhimurium (n = 18), and Enteritidis (n = 11).

	Blood	Liver	Spleen	Rectal content	Total	Numbers (%) of rats carrying the serotype
Salmonella Dublin	0	5	6	0	11	8 (17.3%)
Salmonella II:42:r:-	2	6	2	10	20	18 (39.1%)
Salmonella Kapemba	2	6	6	3	17	12 (26.0%)
Salmonella Mikawasima	0	0	1	1	2	2 (4.3%)
Salmonella Orion	0	0	0	1	1	1 (2.17%)
Salmonella Typhimurium	1	6	4	1	12	10 (21.7%)
Salmonella Weltevreden	0	5	2	3	10	10 (21.7%)
Total	5	28	21	19	73	46

Table 2. Serotypes of rats (n = 46) according to specimen. First isolate per serotype per specimen. A total of 73 single non-duplicate *Salmonella* serotype isolates were recovered from 46 rats; multiple serotypes were observed in 13 rats. Specimens listed were sampled for each rat, except for blood which was sampled for 218/566 (38.5%) rats.

https://doi.org/10.1371/journal.pntd.0010740.t002

Other serotypes were Typhi (n = 6), Paratyphi C (n = 1), II.42:r:- (n = 1), and monophasic I.6,7:y:- (n = 1). Median age of patients infected with iNTS (n = 34) was 20.5 months (3 days-48 years), 81.8% (n = 27/33) were less than 5 years old. Male-to-female ratio was 0.83.

3.4 S. Typhimurium isolates

Results of MLVA and MLST typing of *S*. Typhimurium isolates from rats (12 isolates from 10 rats) and humans (18 isolates from 18 patients available for MLVA typing, 16 isolates from 16 patients available for MLST typing) are listed in Table 3. Fig 2A and 2B depicts the clustering of those 32 isolates through a minimum spanning tree on cgMLST.

The 18 human *S*. Typhimurium isolates comprised nine different MLVA types and all 16 sequenced isolates belonged to MLST ST313. Among the 12 *S*. Typhimurium isolates recovered from 10 rats, four isolates (from three rats) belonged to ST313. The rat ST313 isolates represented 6.5% of *Salmonella*-carrying rats and 0.5% of all rats captured. Rat ST313 isolates belonged to three MLVA types. The remaining eight *S*. Typhimurium isolates from rats belonged to ST19 (four MLVA types, seven rats). The three rats carrying *Salmonella* ST313 did not show signs of illness and had length and mass in line with all the other rats in the study (S2 Table).

Two human-rat clusters were identified among the ST313 isolates, involving the three ST313 carrying rats and 12 out of 16 (75.0%) human isolates (Table 3, Fig 2A and 2B). In the first cluster, two rat isolates (Yate Market, October 2018) grouped with six human isolates obtained between May 2016 and September 2018 (28 months) from patients living between in a perimeter of 3.5 to 3.7 km from the marketplace. In the second cluster, two isolates from one rat (Central Market, October 2016) grouped with six human isolates obtained between August 2016 and May 2017 (8 months) from patients living within a perimeter of 1.2 to 2.9 km from the marketplace.

The ST313 S. Typhimurium human and rat isolates clustered with the ST313 Lineage 2 isolates described by Pulford *et al.* [7], at less than 50 alleles (by the EnteroBase cgMLST scheme), and at more than 200 alleles to the Lineage 1 and Lineage 3 isolates in their study (Fig 3). As to the plasmid repertoire described by Pulford *et al.*[7], there were two distinct profiles: pBT2 and pBT3 (n = 6 humans and n = 1 rat) and pBT3 alone (n = 7 humans and n = 2 rats) (Fig 4). The ten ST19 S. Typhimurium isolates (all eight rats captured at Yaté and Central markets) clustered both by MLVA and HC5 (Fig 5).

3.5 Salmonella Enteritidis isolates

Among 11 human *S*. Enteritidis isolates available, 10 (90.9%) belonged to ST11. When compared on EnteroBase with the genomes available from the study Feasey *et al.* [43], they

Table 3. MLVA types of *Salmonella* Typhimurium isolated from rats and human blood cultures. NA corresponds to a locus on which no allele has been amplified. The two groups of grey shaded cells present MLVA clusters of rat (bold) and human isolates. Two rats (MC071 and MC025) had each two different MLVA types. Code numbers of rats refer to sampling site: AB = Slaughterhouse (Abattoir) Mangobo, MC = Marché Central, MY = Marché Yaté; and specimen type: S = spleen, L = liver, B = blood, ND = no data.

MLVA type	Rats		Humans				
	Code number	ST	Date	Code number	ST	Date	Residence township (distance from rat trapping site)
2-5-8-8-210	-		-	5548/4	313	28/10/16	Mangobo (3.5 km)
2-5-9-8-210	MY302/S MY305/L	313 313	08/10/18 08/10/18	-	-	-	
2-5-11-8-210	- - -		- - -	5232/4 6284/4 6436/4	313 313 313	22/05/16 07/08/18 06/09/18	Kabondo (3.7 km) Kabondo (3.7 km) Tshopo (3.5 km)
2-6-11-8-210	-	-	-	5248/4 5376/4	313 313	26/08/16 10/08/16	Unknown Unknown
2-8-10-8-210		-	-	5809/4 5810/4	313 313	13/05/17 13/05/17	Kisangani (2.9 km) Kisangani (2.9 km)
2-7-10-8-210	MC071/L+S	313	06/10/16	5403/4 5464/4 5598/4 5601/4	313 313 313 313 313	27/08/16 23/09/16 22/11/16 24/11/16	Kabondo (2.6 km) Unknown Kisangani (1.2 km) Kabondo (2.6 km)
2-7-11-8-210	MC071/S	313	06/10/16	-		-	
2-4-11-7-210	-	-	-	7281/4	Not done	13/06/19	
2-7-12-10-210	-	-	-	4930/4	Not done	30/10/15	
2-13-4-3-NA	-	-	-	5835/4	313	31/05/17	
2-NA-12-7-210				5390/4 7279/4 7292/4	313 313 313	16/08/16 10/06/19 14/06/19	
3-12-5-9-311	MC025/L+S	19	15/05/16	-	-	-	
3-12-5-10-311	MC025/L+S	19	15/05/16	-	-	-	
3-15-5-11-311	MY213/S	19	10/06/18	-	-	-	
	MY218/L	19	10/06/18	-	-	-	
	MY365/S	19	16/12/18	-	-	-	
	MY373/L	19	23/12/18	-	-	-	
	MY375/B	19	23/12/18	-	-	-	
2-20-9-7-212	MY300/L	19	08/10/18	-	-	-	

https://doi.org/10.1371/journal.pntd.0010740.t003

clustered at less than 50 alleles to the hierBAPS clade 9, also denominated Central/Eastern African clade (S2 Fig), and carried the expected plasmid pSENT-BT.

3.6 Antimicrobial resistance profile

S. Typhimurium from both rats (all 3) and humans (14 out of 18) displayed MDR, encoding for resistances to a variable number of families of antibiotics. The most frequent profile consisted of AMR determinants to beta-lactams (bla_{TEM-1}), aminoglycosides (*strA*, *strB*, *ant*(3')-*Ia*, *aac*(3')-*IId*), sulfonamides (*sul1*, *sul2*), trimethoprim (*dfrA1*), tetracyclines (*tetB*), chloramphenicol (*catA10*). Among human isolates, occasional decreased ciprofloxacin susceptibility and resistance to azithromycin and ceftriaxone were also observed (Table 4, S3 Table). In contrast, S. Typhimurium ST19 isolates from rats and all other serotypes from rats were susceptible to all antibiotics tested.

ST11 *S*. Enteritidis isolates, all recovered from human samples, encoded for AMR determinants to 5 to 6 families of antibiotics: beta-lactams (*bla*_{TEM-1}), aminoglycosides (*strA*, *strB*), chloramphenicol (*cat2*), sulfonamides (*sul2*), trimethoprim (*dfrA7*) and tetracyclines (*tetA*).



Fig 2. Clustering of S. Typhimurium ST313 human and rat isolates. Minimum spanning tree created using the MSTree V2 component in EnteroBase, based on the allelic differences over the 3002 alleles that constitute the EnteroBase HierCC scheme on cgMLST [41]. The distances between leaves in the tree indicate number of alleles different between genomes. Genomes with common HC5 values are at five or less alleles, strongly indicating epidemic relatedness. Isolates from rats are indicated with an icon and prefix MC and MY (referring to Christ market and Yate market respectively. In panel A, colors are according to cgMLST HC5 values; in panel B, colors are according to MLVA type. Human and rat isolates are grouping in two clusters (MY305/302 and MC071 with each six human isolates. For details of date and place, see Table 3. Rat drawing by Francisca Arévalo from NounProject.com.

4 Discussion

In a sub-Saharan Africa setting endemic for iNTS, the present study showed a *Salmonella* carrier ratio of 8.1% among rats captured in an urban area. Among the serotypes recovered, there



Fig 3. Clustering of S. Typhimurium ST313 human and rat isolates with ST313 Lineage 2 isolates described by Pulford et al. [7]. Minimum spanning tree (MSTree V2) comparing the S. Typhimurium ST313 genomes analyzed by Pulford et al. [7] in 2021 and the genomes in this study. The ST313 genomes in this study (circled in red, filled in white) cluster together with the Lineage 2 strains in the study by Pulford *et al.* [7], colored in green).

was invasive S. Typhimurium ST313 in 3/46 Salmonella-carrying rats which clustered with 13/ 16 human isolates.

4.1 Rats as reservoirs of non-Typhoidal Salmonella

Rodents-in particular rats-are well-known reservoirs of zoonotic pathogens including *Salmo-nella* [26]. Prevalence of *Salmonella* in rats has mainly been assessed in high-income countries and varied widely: in the natural environment, prevalence was consistently low (0 to 1%) [26,44,45] but is was higher in the proximity of human settlements (5–20%) such as cities and



Fig 4. Plasmid profiles of S. Typhimurium ST313 human and rat isolates according to the plasmid profiles of ST313 Lineage 2 isolates described by Pulford *et al.* **[7]. Minimum spanning tree (MSTree V2) comparing all** *S.* **Typhimurium ST313 in this study. 58 alleles separate the two mixed clusters (human and rats isolates) with different plasmid content: in orange, isolates harboring plasmids pBT2 and pBT3; in green, isolates harboring pBT3, only. Rat drawing by Francisca Arévalo from NounProject.com.**

https://doi.org/10.1371/journal.pntd.0010740.g004



Fig 5. Clustering of S. Typhimurium ST19 rat isolates according to HC5-values (cgMLST V2 + HierCC V1, upper left) and MLVA type (lower right). Minimum spanning tree (MSTree V2) comparing all ST19 in this study. In panel A, colors are according to cgMLST HC5 values; in panel B, colors are according to MLVA type.

layer or pig farms (up to 28%) [29-32,46-50]. Of note, studies also showed a high variation in *Salmonella* prevalence between rats in different cities; 15% in Yokohama (Japan) versus 1% in Chicago (US), 0.34% in Nairobi (Kenya) and 0% in Baghdad (Iraq) [32,51-53]. These differences can partly be ascribed to the urban landscape (presence of food and shelter, sanitary conditions); the highest reported prevalence (49.1%) has been observed in wet marketplaces in Udon Thani City, Thailand [54].

In the above-mentioned studies, *Salmonella* serotypes from commensal rats varied widely but there was a consistent overlap with human epidemiologically relevant serotypes. Examples are *Salmonella* Enteritidis and Infantis from layer farms [26,47,55] and *Salmonella* Weltevreden from rats in marketplaces in Thailand [54]. S. Typhimurium and-to a lesser extent–Enteritidis belong to the most observed serotypes [29,31,32,47,48,52,54,56]. Molecular typing allowing for sequence typing of the non-Typhoidal *Salmonella* was not performed so far. The aforementioned study in Japan showed that *S*. Typhimurium isolates obtained from urban rats

Table 4. Antimicrobial resistance profiles of Salmonella Typhimurium isolates from rats and humans (blood cultures). Data present numbers (%) of resistant iso-
lates. Only the first isolate per serotype and rat is listed. MDR: multidrug resistant, *i.e.* resistant to ampicillin, trimethoprim-sulfamethoxazole and chloramphenicol.DCS = decreased ciprofloxacin susceptibility, *i.e.* ciprofloxacin MIC-value > 0.064 mg/L and < 1 mg/L.

		Humans		
Resistance to:	Salmonella Typhimurium ST 313 (N = 3)	Salmonella Typhimurium ST 19 (N = 7)	Other Salmonella (N = 52)	Salmonella Typhimurium (N = 18)
Ampicillin	3	0	0	18
Trimethoprim-sulfamethoxazole	3	0	0	18
Chloramphenicol	3	0	0	14
Multidrug resistant (MDR)	3 (100.0%)	0 (0.0%)	0 (0.0%)	14 (77.8%)
Ceftriaxone	0	0	0	1
Azithromycin	0	0	0	4
Decreased ciprofloxacin susceptibility (DCS)	0	0	0	3
MDR + DCS	0	0	0	1

https://doi.org/10.1371/journal.pntd.0010740.t004

belonged to the DT104 phage type, *i.e.*, the world-wide distributed zoonotic *S*. Typhimurium causing diarrhea in humans [32].

4.2 Rats carried S. Typhimurium ST313, the predominant S. Typhimurium clone in Africa

The 8.1% *Salmonella* prevalence among rats in the present study fits into the prevalence ranges described above. Among the serotypes recovered, the main finding was the presence of *S*. Typhimurium ST313 in three rats. By MLVA as well as by comparison of genomes in Entero-Base, the ST313 isolates grouped in two clusters comprising each six human ST313 isolates recovered from blood cultures. ST313 is the main pathovariant of iNTS in Africa [7]. Both rat and human isolates clustered together with what Pulford *et al.* describe as Lineage 2, *i.e.* the most common and recent *S*. Typhimurium ST313 lineage in the African continent [7]. Close genetic relation was also supported by the identification of the pSLT-BT plasmid, encoding for the different AMR genes that most ST313 isolates from rats and humans in the present study harbored, particularly the *cat1* gene encoding for chloramphenicol resistance, which is hallmark for Lineage 2 [7].

4.3 Finding of *Salmonella* Typhimurium ST313 in rats: laboratory animal models

So far, reservoir and transmission of iNTS including ST313 were hypothesized to be human, based on the genetic signature [4,5,22] and household, case-control, and environmental studies [15,16,20].

Finding ST313 in rats indicates that ST313 is not restricted to the human host. Previous laboratory animal models showed that ST313 was able to cause invasive infections in chickens, mice, and rhesus macaque infection models [6,57–59]. Findings differed according to the challenging strain and host animal but overall ST313 caused, in comparison with ST19, a more rapid onset of bacteremia and infestation of liver and spleen combined with less diarrhea and colonization of the intestinal tract. The fact that in the present study ST313 in rats was exclusively recovered from deep organs may point to an invasive disease too, but none of the rats captured was visibly sick and their length and mass were in line with the other rats. Moreover, although *S*. Typhimurium strains can cause deadly infections, asymptomatic infections of liver and spleen with intermittent fecal shedding can occur too [29,60,61] and may even constitute the majority of infections [62].

4.4 Attribution of rats to Salmonella transmission

Despite being acknowledged as a reservoir for *Salmonella*, there is only circumstantial and anecdotal evidence as to the attribution of rats to the transmission of *Salmonella* [29,61,63]. Rather than being an original reservoir for *Salmonella*, rats are considered to be a "sponge" reflecting the environment they are living in [26,45,51]. This has been demonstrated in layer farms, where proportions of *Salmonella* carrying rats were related to the percentage of *Salmonella* contaminated eggs and the intensity of the environmental contamination [47].

Rats shed *Salmonella* for 2–4 months after infection and have up to 40 droppings a day [29]; *Salmonella* survives long (up to 86 days) in rat droppings [30]. In this way, rats contaminate the layer hen's food and thereby maintain and even amplify *Salmonella* infection in the food production chain [26,47,55]. Conducive to this is the social life of rats (allowing transmission among rats [64] and the long (> 1 year) environmental survival of *Salmonella* [65], although ST313 may have lower environmental persistence compared to the diarrhea-causing

ST19 [11]. In addition, rat carcasses may also constitute a *Salmonella* reservoir on which other animals (stray cats, lizards, cockroaches, ants, etc.) can feed and further contribute to the spread of *Salmonella* [26].

A similar mechanism of indirect ("sponge") reservoir may be postulated for wet markets. In the present study, this is supported by the observation that the marketplaces where ST313 carrying rats were captured (Central Market and Yaté market) were those with the highest rat density. Further, the market setting in a low resource setting such as Kisangani offers exposure of food and shelter in the absence of pest control. By consequence, these are high density populations of rats which was demonstrated by the high trapping success in the present study (one third of non-pre-baited traps) [45]. The present rat population consisted of nearly 90% sexually mature rats versus 63.9% in a rat population in urban Canada; also, length and mass were higher in Kisangani versus Canada (medians of 22.1 cm and 280 g versus 17.5 cm and 142.2 g respectively). This means that the Kisangani rats were probably older, which may increase the probability of rat associated zoonosis [45].

At the human side, socioeconomic factors fueling transmission in Kisangani are numerous: poor sanitary conditions, crowding, overnight stay of vendors, unsafe drinking water and cooking practices. The marketplaces sampled in the present study are easy to reach by the Kisangani population, either by walking or by motorcycle, and human mobility is compatible with the geographical perimeters of the rat-human clusters in the present study. Of note, rats can move too: urban rats are territorial [32,45] but may walk up to 3 km and further in one night [66,67]

4.5 The present findings point to the evidence of an animal reservoir for ST313

As only 6.5% of *Salmonella* carrying rats and 0.5% of all rats captured harbored ST313, it is unlikely that rats constitute a major or preferred reservoir of ST313. However, the rat ST313 isolates clustered with the majority (75.0%) of human isolates, pointing to a close interaction between both. Moreover, the occurrence of ST313 in rat populations postulates also other animals as potential reservoirs of ST313. Previous household studies assessing the iNTS reservoirs in sub-Saharan Africa sampled livestock and poultry but no indoor rodents [15,20,68]. Recently, two studies assessed the meat pathway as a potential source of *Salmonella*. A slaughterhouse study from Kenya and Malawi found a 12.7% *Salmonella* prevalence in pig carcasses but failed to detect iNTS clades [66]. A study from East Africa showed that S. Enteritidis ST11 was present in the meat pathway and clustered with human isolates. Other non-typhoidal *Salmonella* serotypes were observed in the meat pathway too, but not ST313 [23]. Furthermore, a study in DR Congo showed a direct association of rainfall with iNTS, pointing to a possible waterborne environmental reservoir [69].

The potential existence of an environmental reservoir and transmission of iNTS may have an impact on ongoing efforts on iNTS vaccine development, most of which target *S*. Typhimurium [70] and put more emphasis on water, sanitation, and hygiene as control measures. Moreover, ongoing deforestation and climate change are expected to favor both rodent populations and salmonellosis [65,71] with a potential increasing impact of the rat-*Salmonella* reservoir and transmission.

4.6 Other Salmonella serotypes and antimicrobial resistance

The other serotypes obtained from rats mainly comprised II:42:r:-, Kapemba, Weltevreden and Dublin. They were susceptible to all antibiotics tested (= pan-susceptible), as was also the case for the *S*. Typhimurium ST19 isolates. This pan-susceptibility might be explained by the

fact that the non-ST313 *Salmonella* isolates most probably represent animal-confined isolates. Agriculture and livestock raising in the Kisangani area are mainly organized as small-scale subsistence farming with expected a lesser use of antibiotics compared to human medicine. A similar observation was noted in the aforementioned study about *Salmonella* in pigs in Kenya and Malawi: most isolates were susceptible to all antibiotics tested, in line with presumed low antibiotic exposure [72].

Salmonella Dublin is host-adapted to cattle but has been recognized as iNTS too in Mali [73]; it has not yet been detected as part of the microbiological surveillance in DR Congo [22,71]. *Salmonella* Weltevreden is a major cause of intestinal infections in Asia and was cultured from rats in marketplaces in Thailand and India in different environments as poultry farm, feed store and around residential quarters [48,54]. To our knowledge, *Salmonella* sero-types II:42:r:-, Kapemba, Mikawasima and Orion so far have not yet been described in rats.

4.7 Limitations and strengths

A limitation of the present study was the fact that only outdoor sampling was conducted. Indoor sampling would have been a valuable adjunct but in view of the public perception and acceptability, it was presently not done. Likewise, traps were not pre-baited whereas doing so could have increased trapping success as rats are neophobic [45,67]. Trapping success was however high and in line with a previous study [25]. Last, the molecular subtyping alone (as presently used) cannot provide information about the pathways of transmission (rat to human or vice-versa) [13].

Among the strengths were the multiple sampling sites which precluded rat territorial bias [74]. Further, trapping was not impacted by non-target species and, thanks to excellent communication with the city and market authorities, there was no tripping or stealing by the public [74]. The present sample size (n = 566) was large compared to other studies [48,51,72] and human clinical isolates from the same period and area were available for comparison. In addition, we collected physical data of the rats (which may relate to the presence of zoonosis [74] and sampled year-round, precluding seasonal variation. The co-located laboratory facilities allowed for expertized staff and short and regular transport. Deep organs were sampled, and they accounted for three quarters of the non-duplicate serotype isolates. If only stool/rectal swab or content would have been sampled (as done in many studies [32,45,54,75]), carriage would have been as low as 2.6% (15 rats) and ST313 would have been missed. Furthermore, as confirmed here, rats may carry more than one *Salmonella* serotypes [44,45], which is the reason why we processed multiple colonies per sample.

Further research should confirm and extend the present findings to other iNTS-endemic areas. Field sampling should include wet marketplaces but also indoor rodent sampling in households of iNTS index patients. In conjunction with rat studies, market retail products should be assessed for contamination with iNTS [76]. Laboratory experiments should establish the pathophysiology of iNTS in rats in particular intestinal colonization and fecal shedding. At a larger scope, further environmental research into the reservoir of iNTS should be conducted [6,21,23]. Apart from rats, small rodents such as mice should be assessed, as well as lizards and geckos which are notable carriers of *Salmonella* too [77]. Food and water should be further assessed as a vehicle for transmission, including fork-to-farm studies along the meat pathway [23,78], crops and green leaves [13,69]. Source attribution studies such as case-control and outbreak investigation studies should complement the picture of iNTS transmission [13,78].

In conclusion, the present study provided evidence of rats as carriers of *S*. Typhimurium ST313 clustering with human blood culture isolates in an iNTS endemic area in sub-Saharan Africa.

Supporting information

S1 Table. ENA depository genomes. (DOCX)

S2 Table. Morphometric parameters of rats (*Rattus rattus* and *Rattus norvegicus*) related to the *Salmonella* carriage status in Kisangani 2016–2018). (DOCX)

S3 Table. Antibiotic susceptibility data of human and rat *Salmonella* isolates. (DOCX)

S1 Fig. Timeline of human and rat sampling. (DOCX)

S2 Fig. Clustering of human Salmonella Enteritidis isolates with ST11 Salmonella Enteritidis isolates.

(DOCX)

Acknowledgments

We would like to thank the technical staff participating in the fieldwork, dissection, and laboratory analysis, in particular Brigitte Mapendo and Rémy Bongolu.

Author Contributions

Conceptualization: Dadi Falay, Chris Van Geet, Dauly Ngbonda, Jan Jacobs.

Data curation: Dadi Falay, Liselotte Hardy, Octavie Lunguya, Maria Pardos de la Gandara, Jan Jacobs.

Formal analysis: Dadi Falay, Maria Pardos de la Gandara, Jan Jacobs.

Funding acquisition: Chris Van Geet, Dauly Ngbonda, Jan Jacobs.

- **Investigation:** Dadi Falay, Wesley Mattheus, Pionus Katuala, Maria Pardos de la Gandara, Jan Jacobs.
- Project administration: Dadi Falay, Jan Jacobs.

Resources: Chris Van Geet, Pionus Katuala, Dauly Ngbonda, François-Xavier Weill.

- Supervision: Liselotte Hardy, Octavie Lunguya, Chris Van Geet, Erik Verheyen, Dauly Ngbonda, François-Xavier Weill, Jan Jacobs.
- Visualization: Dadi Falay, Liselotte Hardy, Maria Pardos de la Gandara, Jan Jacobs.

Writing - original draft: Dadi Falay, Maria Pardos de la Gandara, Jan Jacobs.

Writing – review & editing: Liselotte Hardy, Jacques Tanzito, Octavie Lunguya, Edmonde Bonebe, Marjan Peeters, Wesley Mattheus, Chris Van Geet, Erik Verheyen, Dudu Akaibe, Pionus Katuala, Dauly Ngbonda, François-Xavier Weill, Maria Pardos de la Gandara.

References

 Stanaway JD, Parisi A, Sarkar K, Blacker BF, Reiner RC, Hay SI, et al. The global burden of non-typhoidal salmonella invasive disease: a systematic analysis for the Global Burden of Disease Study 2017. The Lancet Infectious Diseases. 2019; 19: 1312–1324. https://doi.org/10.1016/S1473-3099(19)30418-9 PMID: 31562022

- Feasey NA, Dougan G, Kingsley RA, Heyderman RS, Gordon MA. Invasive non-typhoidal salmonella disease: An emerging and neglected tropical disease in Africa. The Lancet. 2012; 379: 2489–2499. https://doi.org/10.1016/S0140-6736(11)61752-2 PMID: 22587967
- Gordon MA. Invasive nontyphoidal Salmonella disease: Epidemiology, pathogenesis and diagnosis. Current Opinion in Infectious Diseases. 2011. pp. 484–489. <u>https://doi.org/10.1097/QCO.</u> 0b013e32834a9980 PMID: 21844803
- Kingsley RA, Msefula CL, Thomson NR, Kariuki S, Holt KE, Gordon MA, et al. Epidemic multiple drug resistant Salmonella Typhimurium causing invasive disease in sub-Saharan Africa have a distinct genotype. Genome Research. 2009; 19: 2279–2287. https://doi.org/10.1101/gr.091017.109 PMID: 19901036
- Canals R, Hammarlöf DL, Kröger C, Owen S v., Fong WY, Lacharme-Lora L, et al. Adding function to the genome of African Salmonella Typhimurium ST313 strain D23580. PLoS Biology. 2019; 17. https://doi.org/10.1371/journal.pbio.3000059 PMID: 30645593
- Okoro CK, Barquist L, Connor TR, Harris SR, Clare S, Stevens MP, et al. Signatures of Adaptation in Human Invasive Salmonella Typhimurium ST313 Populations from Sub-Saharan Africa. PLoS Neglected Tropical Diseases. 2015; 9. https://doi.org/10.1371/journal.pntd.0003611 PMID: 25803844
- Pulford C v., Perez-Sepulveda BM, Canals R, Bevington JA, Bengtsson RJ, Wenner N, et al. Stepwise evolution of Salmonella Typhimurium ST313 causing bloodstream infection in Africa. Nature Microbiology. 2021; 6: 327–338. https://doi.org/10.1038/s41564-020-00836-1 PMID: 33349664
- Lokken KL, Walker GT, Tsolis RM. Disseminated infections with antibiotic-resistant non-typhoidal Salmonella strains: Contributions of host and pathogen factors. Pathogens and Disease. Oxford University Press; 2016. https://doi.org/10.1093/femspd/ftw103 PMID: 27765795
- The European Union One Health 2020 Zoonoses Report. EFSA Journal. 2021; 19. <u>https://doi.org/10.2903/J.EFSA.2021.6971</u>
- Crump JA, Sjölund-Karlsson M, Gordon MA, Parry CM. Epidemiology, clinical presentation, laboratory diagnosis, antimicrobial resistance, and antimicrobial management of invasive Salmonella infections. Clinical Microbiology Reviews. American Society for Microbiology; 2015. pp. 901–937. https://doi.org/ 10.1128/CMR.00002-15 PMID: 26180063
- Gilchrist JJ, MacLennan CA. Invasive Nontyphoidal Salmonella Disease in Africa. EcoSal Plus. 2019; 8. https://doi.org/10.1128/ecosalplus.ESP-0007-2018 PMID: 30657108
- Tack B, Vanaenrode J, Verbakel JY, Toelen J, Jacobs J. Invasive non-typhoidal Salmonella infections in sub-Saharan Africa: a systematic review on antimicrobial resistance and treatment. BMC Medicine. 2020; 18. https://doi.org/10.1186/s12916-020-01652-4 PMID: 32677939
- Pires SM, Vieira AR, Hald T, Cole D. Source attribution of human salmonellosis: An overview of methods and estimates. Foodborne Pathogens and Disease. 2014; 11: 667–676. <u>https://doi.org/10.1089/fpd.2014.1744</u> PMID: 24885917
- Majowicz SE, Musto J, Scallan E, Angulo FJ, Kirk M, O'Brien SJ, et al. The global burden of nontyphoidal salmonella gastroenteritis. Clinical Infectious Diseases. 2010; 50: 882–889. https://doi.org/10.1086/ 650733 PMID: 20158401
- Kariuki S, Revathi G, Kariuki N, Kiiru J, Mwituria J, Muyodi J, et al. Invasive multidrug-resistant nontyphoidal Salmonella infections in Africa: Zoonotic or anthroponotic transmission? Journal of Medical Microbiology. 2006; 55: 585–591. https://doi.org/10.1099/jmm.0.46375-0 PMID: 16585646
- 16. Kariuki S, Mbae C, van Puyvelde S, Onsare R, Kavai S, Wairimu C, et al. High relatedness of invasive multi-drug resistant non-typhoidal Salmonella genotypes among patients and asymptomatic carriers in endemic informal settlements in Kenya. PLOS Neglected Tropical Diseases. 2020; 14: e0008440. https://doi.org/10.1371/journal.pntd.0008440 PMID: 32745137
- 17. Morpeth SC, Ramadhani HO, Crump JA. Invasive non-Typhi Salmonella disease in Africa. Clinical Infectious Diseases. 2009. pp. 606–611. https://doi.org/10.1086/603553 PMID: 19591599
- Phoba MF, Barbé B, Ley B, van Puyvelde S, Post A, Mattheus W, et al. High genetic similarity between non-typhoidal salmonella isolated from paired blood and stool samples of children in the Democratic Republic of the Congo. PLoS Neglected Tropical Diseases. 2020; 14: 1–15. https://doi.org/10.1371/ journal.pntd.0008377 PMID: 32614856
- Tack B, Phoba MF, Barbé B, Kalonji LM, Hardy L, van Puyvelde S, et al. Non-typhoidal salmonella bloodstream infections in Kisantu, DR Congo: Emergence of O5-negative salmonella typhimurium and extensive drug resistance. PLoS Neglected Tropical Diseases. 2020; 14: 1–22. <u>https://doi.org/10.1371/journal.pntd.0008121</u> PMID: 32240161
- Post AS, Diallo SN, Guiraud I, Lompo P, Tahita MC, Maltha J, et al. Supporting evidence for a human reservoir of invasive non-Typhoidal Salmonella from household samples in Burkina Faso. PLoS Neglected Tropical Diseases. 2019; 13. https://doi.org/10.1371/journal.pntd.0007782 PMID: 31609964

- Mather AE, Vaughan TG, French NP. Molecular approaches to understanding transmission and source attribution in nontyphoidal salmonella and their application in Africa. Clinical Infectious Diseases. 2015; 61: S259–S265. https://doi.org/10.1093/cid/civ727 PMID: 26449940
- Okoro CK, Kingsley RA, Quail MA, Kankwatira AM, Feasey NA, Parkhill J, et al. High-resolution single nucleotide polymorphism analysis distinguishes recrudescence and reinfection in recurrent invasive nontyphoidal salmonella typhimurium disease. Clinical Infectious Diseases. 2012; 54: 955–963. <u>https:// doi.org/10.1093/cid/cir1032</u> PMID: 22318974
- 23. Crump JA, Thomas KM, Benschop J, Knox MA, Wilkinson DA, Midwinter AC, et al. Investigating the Meat Pathway as a Source of Human Nontyphoidal Salmonella Bloodstream Infections and Diarrhea in East Africa. Clin Infect Dis. 2021; 73: e1570–e1578. https://doi.org/10.1093/cid/ciaa1153 PMID: 32777036
- 24. Falay D, Kuijpers LMF, Phoba MF, de Boeck H, Lunguya O, Vakaniaki E, et al. Microbiological, clinical and molecular findings of non-typhoidal Salmonella bloodstream infections associated with malaria, Oriental Province, Democratic Republic of the Congo. BMC Infectious Diseases. 2016; 16. <u>https://doi.org/10.1186/s12879-016-1604-1 PMID: 27286886</u>
- 25. Laudisoit A, Falay D, Amundala N, Akaibe D, de Bellocq JG, van Houtte N, et al. High prevalence of Rickettsia typhi and Bartonella species in rats and fleas, Kisangani, Democratic Republic of the Congo. American Journal of Tropical Medicine and Hygiene. 2014; 90: 463–468. <u>https://doi.org/10.4269/ajtmh. 13-0216</u> PMID: 24445202
- Meerburg BG, Kijlstra A. Role of rodents in transmission of Salmonella and Campylobacter. Journal of the Science of Food and Agriculture. 2007. pp. 2774–2781. https://doi.org/10.1002/jsfa.3004
- Sikes RS, Mammalogists the AC and UC of the AS of. 2016 Guidelines of the American Society of Mammalogists for the use of wild mammals in research and education. Journal of Mammalogy. 2016; 97: 663–688. https://doi.org/10.1093/jmammal/gyw078 PMID: 29692469
- CLSI. M39-A4 Analysis and presentation of cumulative antimicrobial susceptibility test data; Approved Guideline—Fourth Edition. 2014. Available: http://shop.clsi.org/c.1253739/site/Sample_pdf/M39A4_ sample.pdf
- Healing TD. Salmonella in rodents: a risk to man? Communicable Disease Report (Lond Engl Rev). 1991; 1: 1223–1227.
- Hilton AC, Willis RJ, Hickie SJ. Isolation of Salmonella from urban wild brown rats (Rattus norvegicus) in the West Midlands, UK. International Journal of Environmental Health Research. 2002; 12: 163–168. https://doi.org/10.1080/09603120220129328 PMID: 12396533
- Seguin B, Boucaud-maître Y, Quenin B, Lorgue B. Bilan épidémiologique d'un échantillon de 91 rats (Rattus norvegicus) capturés dans les égouts de Lyon. Zentralbl Bakteriol Orig A. 1986; 539–546.
- 32. Yokoyama E, Maruyama S, Kabeya H, Hara S, Sata S, Kuroki T, et al. Prevalence and genetic properties of Salmonella enterica serovar Typhimurium definitive phage type 104 isolated from Rattus norvegicus and Rattus rattus house rats in Yokohama City, Japan. Applied and Environmental Microbiology. 2007; 73: 2624–2630. https://doi.org/10.1128/AEM.02465-06 PMID: 17308195
- 33. Ministère de Plan et Suivi de la Mise en oeuvre de la Révolution de la Modernité (MPSMRM), Ministère de la Santé Publique (MSP), ICF International. Enquête démographique et de santé en République Démocratique du Congo 2013–2014. Rockville, Maryland, USA; 2014. Available: www.DHSprogram. com.
- 34. Kahle D, Wickham H. ggmap: Spatial Visualization with ggplot2. The R Journal. 2013; 5: 144–161.
- 35. Mbuyi-Kalonji L, Barbé B, Nkoji G, Madinga J, Roucher C, Linsuke S, et al. Non-typhoidal Salmonella intestinal carriage in a Schistosoma mansoni endemic community in a rural area of the Democratic Republic of Congo. PLoS Neglected Tropical Diseases. 2020; 14. <u>https://doi.org/10.1371/journal.pntd.</u> 0007875 PMID: 32084128
- CLSI. Performance Standards for Antimicrobial Susceptibility Testing. 31th ed. CLSI supplement M100. 31st ed. Clinical and Laboratory Standards Institute; 2021.
- Nikiema MEM, de La Gandara MP, Compaore KAM, Ba AK, Soro KD, Nikiema PA, et al. Contamination of street food with multidrug-resistant Salmonella, in Ouagadougou, Burkina Faso. PLoS One. 2021; 16. https://doi.org/10.1371/journal.pone.0253312 PMID: 34138936
- Dimovski K, Cao H, Wijburg OLC, Strugnell RA, Mantena RK, Whipp M, et al. Analysis of Salmonella enterica serovar Typhimurium variable-number tandem-repeat data for public health investigation based on measured mutation rates and whole-genome sequence comparisons. J Bacteriol. 2014; 196: 3036–3044. https://doi.org/10.1128/JB.01820-14 PMID: 24957617
- Kidgell C, Reichard U, Wain J, Linz B, Torpdahl M, Dougan G, et al. Salmonella typhi, the causative agent of typhoid fever, is approximately 50,000 years old. Infection, Genetics and Evolution. 2002; 2: 39–45. https://doi.org/10.1016/s1567-1348(02)00089-8 PMID: 12797999

- Achtman M, Wain J, Weill FX, Nair S, Zhou Z, Sangal V, et al. Multilocus sequence typing as a replacement for serotyping in Salmonella enterica. PLoS Pathog. 2012; 8. https://doi.org/10.1371/journal.ppat. 1002776 PMID: 22737074
- Zhou Z, Charlesworth J, Achtman M. HierCC: A multi-level clustering scheme for population assignments based on core genome MLST. Bioinformatics. 2021; 37: 0–0. <u>https://doi.org/10.1093/bioinformatics/btab234</u> PMID: 33823553
- 42. Robinson E, Travanut M, Fabre L, Larréché S, Ramelli L, Pascal L, et al. Outbreak of Salmonella Newport associated with internationally distributed raw goats' milk cheese, france, 2018. Epidemiology and Infection. 2020. https://doi.org/10.1017/S0950268820000904 PMID: 32364094
- 43. Feasey NA, Hadfield J, Keddy KH, Dallman TJ, Jacobs J, Deng X, et al. Distinct Salmonella Enteritidis lineages associated with enterocolitis in high-income settings and invasive disease in low-income settings. Nature Genetics. 2016; 48: 1211–1217. https://doi.org/10.1038/ng.3644 PMID: 27548315
- Skarżyńska M, Zajac, M, Kamińska E, Bomba A, Żmudzki J, Jabłoński A, et al. Salmonella and antimicrobial resistance in wild rodents—true or false threat? Pathogens. 2020; 9: 1–13. <u>https://doi.org/10.3390/pathogens9090771</u> PMID: 32967245
- 45. Himsworth CG, Parsons KL, Jardine C, Patrick DM. Rats, cities, people, and pathogens: A systematic review and narrative synthesis of literature regarding the ecology of rat-associated zoonoses in urban centers. Vector-Borne and Zoonotic Diseases. 2013. pp. 349–359. <u>https://doi.org/10.1089/vbz.2012</u>. 1195 PMID: 23590323
- 46. Pasmans F, van Immerseel F, Hermans K, Heyndrickx M, Collard JM, Ducatelle R, et al. Assessment of Virulence of Pigeon Isolates of Salmonella enterica subsp. enterica Serovar Typhimurium Variant Copenhagen for Humans. Journal of Clinical Microbiology. 2004; 42: 2000–2002. https://doi.org/10. 1128/JCM.42.5.2000-2002.2004 PMID: 15131161
- Lapuz R, Tani H, Sasai K, Shirota K, Katoh H, Baba E. The role of roof rats (Rattus rattus) in the spread of Salmonella Enteritidis and S. Infantis contamination in layer farms in eastern Japan. Epidemiology and Infection. 2008; 136: 1235–1243. https://doi.org/10.1017/S095026880700948X PMID: 17988423
- Singh Suresh P, Sethi M S, Sharma V D. The occurence of Salmonellae in rodent, shrew, cockroch and ant. InstJZoon. 1980; 7: 58–61.
- Antoniou M, Psaroulaki A, Toumazos P, Mazeris A, Ioannou I, Papaprodromou M, et al. Rats as Indicators of the Presence and Dispersal of Pathogens in Cyprus: Ectoparasites, Parasitic Helminths, Enteric Bacteria, and Encephalomyocarditis Virus. 2010.
- Andrés-Barranco S, Vico JP, Garrido V, Samper S, Herrera-León S, de Frutos C, et al. Role of wild bird and rodents in the epidemiology of subclinical salmonellosis in finishing pigs. Foodborne Pathogens and Disease. 2014; 11: 689–697. https://doi.org/10.1089/fpd.2014.1755 PMID: 24927384
- Murray MH, Fidino M, Fyffe R, Byers KA, Pettengill JB, Sondgeroth KS, et al. City sanitation and socioeconomics predict rat zoonotic infection across diverse neighbourhoods. Zoonoses and Public Health. 2020; 67: 673–683. https://doi.org/10.1111/zph.12748 PMID: 32583624
- Gakuya F M, Kyule M N, Gathura P B, Kariuki S. Antimicrobial resistance of bacterial organisms isolated from rats. East African Medical Journal. 2001; 78: 646–649. https://doi.org/10.4314/eamj.v78i12. 8934 PMID: 12199446
- Ayyal NM, Abbas ZA, Karim AJ, Abbas ZM, Al-Salihi KA, Khalaf JM, et al. Bacterial isolation from internal organs of rats (Rattus rattus) captured in Baghdad city of Iraq. Veterinary World. 2019; 12: 119–125. https://doi.org/10.14202/vetworld.2019.119-125 PMID: 30936664
- 54. Alexis Ribas, Weerachai Saijuntha, Takeshi Agatsuma, Veronika Prantlova, Srisupaph Poonlaphdecha. Rodents as a Source of Salmonella Contamination in Wet Markets in Thailand. Vector-borne and zoonotic diseases. 2016; 16: 537–540. https://doi.org/10.1089/vbz.2015.1894 PMID: 27400325
- Umali D v., Lapuz RRSP, Suzuki T, Shirota K, Katoh H. Transmission and shedding patterns of Salmonella in naturally infected captive wild roof rats (rattus rattus) from a Salmonella-contaminated layer farm. Avian Diseases. 2012; 56: 288–294. <u>https://doi.org/10.1637/9911-090411-Reg.1</u> PMID: 22856184
- Runge M, von Keyserlingk M, Braune S, Becker D, Plenge-Bönig A, Freise JF, et al. Distribution of rodenticide resistance and zoonotic pathogens in Norway rats in Lower Saxony and Hamburg, Germany. Pest Management Science. 2013; 69: 403–408. <u>https://doi.org/10.1002/ps.3369</u> PMID: 22888034
- 57. Parsons BN, Humphrey S, Salisbury AM, Mikoleit J, Hinton JCD, Gordon MA, et al. Invasive Non-Typhoidal Salmonella Typhimurium ST313 Are Not Host-Restricted and Have an Invasive Phenotype in Experimentally Infected Chickens. PLoS Neglected Tropical Diseases. 2013; 7. <u>https://doi.org/10.1371/journal.pntd.0002487 PMID: 24130915</u>

- Yang J, Barrila J, Roland KL, Kilbourne J, Ott CM, Forsyth RJ, et al. Characterization of the invasive, multidrug resistant non-typhoidal Salmonella strain D23580 in a murine model of infection. PLoS Neglected Tropical Diseases. 2015; 9. https://doi.org/10.1371/journal.pntd.0003839 PMID: 26091096
- Ramachandran G, Panda A, Higginson EE, Ateh E, Lipsky MM, Sen S, et al. Virulence of invasive Salmonella Typhimurium ST313 in animal models of infection. PLoS Neglected Tropical Diseases. 2017; 11. https://doi.org/10.1371/journal.pntd.0005697 PMID: 28783750
- 60. Wray C, Wray A, editors. Salmonella in Domestic Animals. London: CABI Publishing; 2000.
- Swanson J Stephen, Snider Cynthia, Braden R Christopher, Boxrud David, Wünschmann Arno, Rudroff Jo Ann, et al. Multidrug-Resistent Salmonella enterica Serotype Typhimurium Associated with Pet Rodents. The New England Jounal of Medicine. 2007; 356: 21–28.
- Hoelzer K, Switt AIM, Wiedmann M. Animal contact as a source of human non-typhoidal salmonellosis. Veterinary Research. 2011; 42: 34. https://doi.org/10.1186/1297-9716-42-34 PMID: 21324103
- Wright G J, Tengelsen A L, Smith E K, Bender B J, Frank K R, Grendon H J, et al. Multidrug-resistant Salmonella Typhimurium in Four Animal Facilities. Emerging Infectious Diseases. 2005; 11: 1235– 1241. https://doi.org/10.3201/eid1108.050111 PMID: 16102313
- 64. Welch H, Ostrolenk M, Bartram MT, Bacteriologists A. Role of Rats in the Spread of Food Poisoning Bacteria of the Salmonella Group*. 1941.
- **65.** Hellberg RS, Chu E. Effects of climate change on the persistence and dispersal of foodborne bacterial pathogens in the outdoor environment: A review. Critical Reviews in Microbiology. 2016; 42: 548–572. https://doi.org/10.3109/1040841X.2014.972335 PMID: 25612827
- Backhans A, Jacobson M, Hansson I, Lebbad M, Lambertz ST, Gammelgård E, et al. Occurrence of pathogens in wild rodents caught on Swedish pig and chicken farms. Epidemiology and Infection. 2012; 141: 1885–1891. https://doi.org/10.1017/S0950268812002609 PMID: 23174339
- Byers KA, Lee MJ, Patrick DM, Himsworth CG. Rats about town: A systematic review of rat movement in urban ecosystems. Frontiers in Ecology and Evolution. Frontiers Media S.A.; 2019. https://doi.org/10. 3389/fevo.2019.00013
- Dione MM, Ikumapayi U, Saha D, Mohammed NI, Adegbola RA, Geerts S, et al. Antimicrobial resistance and virulence genes of non-typhoidal Salmonella isolates in The Gambia and Senegal. 2011.
- 69. Tack B, Vita D, Phoba MF, Mbuyi-Kalonji L, Hardy L, Barbé B, et al. Direct association between rainfall and non-typhoidal Salmonella bloodstream infections in hospital-admitted children in the Democratic Republic of Congo. Scientific Reports. 2021; 11. <u>https://doi.org/10.1038/s41598-021-01030-x</u> PMID: 34732799
- Balasubramanian R, Im J, Lee JS, Jeon HJ, Mogeni OD, Kim JH, et al. The global burden and epidemiology of invasive non-typhoidal Salmonella infections. Human Vaccines and Immunotherapeutics. 2019; 15: 1421–1426. https://doi.org/10.1080/21645515.2018.1504717 PMID: 30081708
- Young HS, Dirzo R, Helgen KM, McCauley DJ, Billeter SA, Kosoy MY, et al. Declines in large wildlife increase landscape-level prevalence of rodent-borne disease in Africa. Proc Natl Acad Sci U S A. 2014; 111: 7036–7041. https://doi.org/10.1073/pnas.1404958111 PMID: 24778215
- 72. Wilson CN, Pulford C v., Akoko J, Sepulveda BP, Predeus A v., Bevington J, et al. Salmonella identified in pigs in kenya and malawi reveals the potential for zoonotic transmission in emerging pork markets. PLoS Neglected Tropical Diseases. 2020; 14: 1–16. <u>https://doi.org/10.1371/journal.pntd.0008796</u> PMID: 33232324
- 73. Tennant SM, Diallo S, Levy H, Livio S, Sow SO, Tapia M, et al. Identification by PCR of non-typhoidal Salmonella enterica serovars associated with invasive infections among febrile patients in Mali. PLoS Neglected Tropical Diseases. 2010; 4: 1–9. https://doi.org/10.1371/journal.pntd.0000621 PMID: 20231882
- 74. Himsworth G Chelsea, Patrick M David, Mak Sunny, Jardine M Claire, Tang Patrick, Weese Scott J. Carriage of Clostridium difficile by Wild Urban Norway Rats (Rattus norvegicus) and Black Rats (Rattus rattus). Applied and Environmental Microbiology. 2014; 80: 1299–1305. <u>https://doi.org/10.1128/AEM.</u> 03609-13 PMID: 24317079
- 75. Nkogwe C, Raletobana J, Stewart-Johnson A, Suepaul S, Adesiyun A. Frequency of detection of Escherichia coli, Salmonella spp., and Campylobacter spp. in the Faeces of Wild Rats (Rattus spp.) in Trinidad and Tobago. Veterinary Medicine International. 2011; 2011. <u>https://doi.org/10.4061/2011/ 686923 PMID: 21547220</u>
- 76. Santos PDM, Widmer KW, Rivera WL. PCR-based detection and serovar identification of Salmonella in retail meat collected from wet markets in Metro Manila, Philippines. PLoS ONE. 2020; 15. <u>https://doi.org/10.1371/journal.pone.0239457</u> PMID: 32997676

- 77. Whiley H, Gardner MG, Ross K. A review of salmonella and squamates (Lizards, snakes and amphisbians): Implications for public health. Pathogens. MDPI AG; 2017. https://doi.org/10.3390/ pathogens6030038 PMID: 28829352
- 78. Thomas KM, de Glanville WA, Barker GC, Benschop J, Buza JJ, Cleaveland S, et al. Prevalence of Campylobacter and Salmonella in African food animals and meat: A systematic review and meta-analysis. Int J Food Microbiol. 2020; 315. https://doi.org/10.1016/j.ijfoodmicro.2019.108382 PMID: 31710971